Author’s response to reviews

Title: What is needed to guide testing for anorectal and pharyngeal Chlamydia trachomatis and Neisseria gonorrhoeae in women and men? Evidence and Opinion

Authors:

Nicole Dukers-Muijrers (nicole.dukers@ggdzn.nl)
Julius Schachter (julius.schachter@ucsf.edu)
Petra Wolffs (p.wolffs@mumc.nl)
Genevieve van Liere (genevieve.vanliere@ggdzn.nl)
Christian Hoebe (christian.hoebe@ggdzn.nl)

Version: 3 Date: 09 Nov 2015

Author’s response to reviews:

Dear Editor,

Thank you for considering our manuscript as potentially acceptable for publication.

You write '[manuscript]...has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Infectious Diseases, once you have carried out some essential revisions suggested by our reviewers'.

Editorial requests:

1. Ensure that there is only one copy of each table and each table legend in the main manuscript: remove the additional copies from the main text and include a single copy after the References
   - Done

2. Upload a clean copy of the manuscript without tracked changes.
   - Done

Further, some very minor corrections were made in the text of the manuscript and in the Tables (version with marked up changes is attached). Please also note that there were no additional reviewer comments (which you refer to).
Kind regards,

Nicole Dukers-Muijrs